Author's response to reviews

Title: Clinical and genetic analyses of three Korean families with hereditary hemorrhagic telangiectasia

Authors:

Mi-Jung Kim (kimmj@gachon.ac.kr)
Seon-Tae Kim (rhinokim2002@hanmail.net)
Hyoung Doo Lee (hdlee@pusan.ac.kr)
Kyu-Yong Lee (kylee@hanyang.ac.kr)
Jiyoung Seo (jyseo@gachon.ac.kr)
Jae-Bom Lee (ljbom@hotmail.com)
Young Jae Lee (leeyj@gachon.ac.kr)
S. Paul Oh (ohp@ufl.edu)

Version: 4 Date: 23 September 2011

Author's response to reviews:

We have revised our manuscript as suggested.
1. "MJ Kim et al" was removed from each page.
2. Several typos and grammatical errors throughout the manuscript were corrected.
3. The order of figures were changed so that they correspond to the order appear in the text: previous Figure 1 to Figure 3; previous Figure 2 to Figure 1; previous Figure 3 to Figure 2. Figure legends were also changed accordingly.
4. Previously labeled as "supplementary material" is now added to the end of manuscript as "Additional files".